Author’s response to reviews

Title: Effects of Panax notoginseng saponins on severe acute pancreatitis through the regulation of mTOR/Akt and caspase-3 signaling pathway by upregulating miR-181b expression in rats

Authors:

Ming-wei Liu (lmw2004210@163.com)
Rui Wei (758437603@qq.com)
Mei-xian Su (Kmsmx24@126.com)
Hui Li (3420220794@qq.com)
Tian-wen Fang (13890801975@163.com)
Wei Zhang (2436730979@qq.com)

Version: 5 Date: 25 Jan 2018

Author’s response to reviews:

The Biomed Central Editorial Team 24 January 2018

BCAM-D-17-00608R4

Effects of Panax notoginseng saponins on severe acute pancreatitis through the regulation of mTOR/Akt and caspase-3 signaling pathway by upregulating miR-181b expression in rats

Ming-wei Liu; Rui Wei; Mei-xian Su; Hui Li; Tian-wen Fang; Wei Zhang

BMC Complementary and Alternative Medicine

Dear Dr Faye Nourollahi,

Thank you for consideration of our manuscript for publication in your journal.

We have reviewed the above manuscript according to your editorial and reviewer’s comments.
Responses to the comments of Reviewer

Editor Comments:

BMC Complementary and Alternative Medicine operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

I have viewed the reports and seen the names of the reviewers

Reviewer reports:

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandreviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.

Improvements to the English language within our manuscript haven't been requested, however, this manuscript has been thoroughly edited by a native English speaker from an editing company.

----------------------

Editorial Policies

----------------------

Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements this will cause a delay whilst the issue is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

In accordance with BioMed Central editorial policies and formatting guidelines, all submissions to BMC Complementary and Alternative Medicine must have a Declarations section which
includes the mandatory sub-sections listed below. Please refer to the journal's Submission Guidelines web page for information regarding the criteria for each sub-section (https://bmccomplementalternmed.biomedcentral.com/).

Where a mandatory section is not relevant to your study design or article type, for example, if your manuscript does not contain any individual persons data, please write "Not applicable" in these sections.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must state that data will not be shared, and provide reasons for this in the manuscript text. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page - http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials.

Declarations
- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information

Adhere to our editorial requirements

According to your editorial requirements, we have read the following information and revised our manuscript as necessary.
In accordance with BioMed Central editorial policies and formatting guidelines, the manuscript have a Declarations section. According to the mandatory sub-sections listed, we removed “Plant material statement” and “Publisher's Note”, added “Authors' Information”.

For the 'Availability of data and materials' section, We have provided information about where the data supporting your findings can be found in revised manuscript.

Best wishes,

Prof Ming-wei Liu